Author's response to reviews

Title: Deprived Children or Deprived Neighbourhoods? A public health approach to the investigation of links between deprivation and injury risk with specific reference to child road safety.

Authors:

Paul J Hewson (phewson@devon.gov.uk)

Version: 5 Date: 10 May 2004

As per your email of May 10th in which you have provisionally accepted the paper, I have made the amendments requested. Specifically, figure and table placement text has been removed, legends have been removed from figures and unnecessary capitalisation in subject headings has been attended to. Also, referees minor comment that Table 2 comment appears twice has been addressed (duplicated comments removed).